Author’s response to reviews

Title: Targeting insulin resistance in type 2 diabetes via immune modulation of cord blood-derived multipotent stem cells (CB-SCs) in stem cell educator therapy: phase I/II clinical trial

Authors:

Yong Zhao (yzhaowhl@yahoo.com)
Zhaoshun Jiang (zhaoshun_jiang@yahoo.cn)
Tingbao Zhao (zhaoqingbao@yahoo.cn)
Mingliang Ye (mingliang_ye@yahoo.cn)
Chengjin Hu (chengjin_hu@yahoo.cn)
Huimin Zhou (zhouhuimin86@yahoo.cn)
Zhaohui Yin (yin.zhaohui@yahoo.cn)
Yana Chen (yanachen@yahoo.cn)
Ye Zhang (zye17@yahoo.cn)
Shanfen Wang (wshanfen@yahoo.cn)
Jie Shen (shenjie342@yahoo.cn)
Hatim Thaker (hatimthaker@yahoo.com)
Summit Jain (summitjain58@yahoo.com)
Yunxiang Li (li_yunxiang@yahoo.cn)
Yalin Diao (yalindiao@yahoo.cn)
Yingjian Chen (yingjian_chen@yahoo.cn)
Xiaoming Sun (xiaomingsun52@yahoo.com)
Mary Beth Fisk (info@tianhecell.com)
Heng Li (hli19@yahoo.cn)

Version: 4 Date: 25 May 2013

Author’s response to reviews: see over
Dear Dr. Denyer:

Please find enclosed the 2nd revision of manuscript (MS: 8902020692081360) “Targeting insulin resistance in type 2 diabetes via immune modulation of cord blood-derived multipotent stem cells (CB-SCs) in stem cell educator therapy: phase I/II clinical trial”.

We agree with your editorial changes. Based on your guiding, we have revised our manuscript. All changes have been highlighted in the revision.

Thank you for your kind consideration of our paper.

Sincerely yours,

Yong